Author's response to reviews

Title: A proof of principle for using adaptive testing in Routine Outcome Monitoring: the efficiency of the Mood and Anxiety Symptoms Questionnaire - Anhedonic Depression CAT

Authors:

Niels Smits (n.smits@psy.vu.nl)
Frans G Zitman (f.g.zitman@lumc.nl)
Pim Cuijpers (p.cuijpers@psy.vu.nl)
Margien E den Hollander (m.e.den_hollander@lumc.nl)
Ingrid VE Calier (i.v.e.carlier@lumc.nl)

Version: 3 Date: 9 January 2012

Author's response to reviews:

Dear Sir/Madam,

I am very glad the paper is accepted! As requested by you, I changed the following:

Major revisions

Abstract: I removed the reference to a citation and a reference to the appendix.

Structure: I changed the discussion section name into "Discussion and Conclusions".

Figures: I changed the first figure mentioned into the first figure.

Minor revisions

Highlighting/tracking: I removed the highlighting from the text. I removed the figure placement text (although two Colons remained).

Tables: The table titles now appear above the tables.

Typography: I removed some typo's and improved some text.

Yours sincerely,

Niels Smits